\documentclass[11pt,reqno]{amsart}
\usepackage[cp1251]{inputenc}
\usepackage[english]{babel}
\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{amsfonts}
\usepackage{graphicx}
\usepackage{xcolor}
\usepackage[colorlinks]{hyperref}


\usepackage{titlesec}
\titlespacing{\section}{0pt}{2.5ex}{1.5ex}
\titlespacing{\subsection}{0pt}{1.5ex}{1ex}
\titlespacing{\subsubsection}{0pt}{1.5ex}{1ex}
\titleformat{\section}{\large\bfseries\centering}{\thesection}{1em}{}
\titleformat{\subsection}[runin]{\bfseries}{\thesubsection.}{0.5em}{}[.\mbox{\ }]
\titleformat{\subsubsection}[runin]{\bfseries}{\thesubsubsection.}{0.4em}{}[.\mbox{\ }]



%%%%%---------- ORCiD
\newcommand\orcidicon[1]{\href{https://orcid.org/#1}{\includegraphics[scale=0.02]{orcid.pdf}}}

\def\udcs{517.987} %Здесь автор определяет УДК своей работы
\def\mscs{37A30, 37A15} %Здесь автор определяет классификаторы AMS своей работы
\setcounter{page}{144}

\newtheorem{lemma}{Lemma}
\newtheorem{theorem}{Theorem}
\newtheorem{definition}{Определение}
\newtheorem{proposition}{Proposition}


\definecolor{myblue}{rgb}{0.03, 0.27, 0.79}

\hypersetup{linkcolor=blue,filecolor=black,urlcolor=blue, citecolor=blue}

\def\logo{\textcolor{blue}{{\bf\Huge S\raisebox{0.2ex}{\hspace{0.55ex}\raisebox{0.05ex}e\hspace{-1.65ex}$\bigcirc$}MR}}}

\def\semrtop
{
	\vbox{
		\noindent\logo \hspace{-3mm}  \parbox{12cm}{     \begin{center}
				{\LARGE \textsf{СИБИРСКИЕ ЭЛЕКТРОННЫЕ}} \\[2mm]
				{\LARGE \textsf{МАТЕМАТИЧЕСКИЕ ИЗВЕСТИЯ}} \\[2mm]
				{\Large \textsf{Siberian Electronic Mathematical Reports}} \\[1mm]
				{\Large\tt{\textcolor{blue}{http://semr.math.nsc.ru}}}\\[1mm]
				\raisebox{1ex}{ISSN 1813-3304 }
				\vspace*{-3mm}
				%     {\small Институт математики им. С.Л.Соболева СО РАН}\\[-1mm]
				%     {\small Sobolev Institute of Mathematics SB RAS}
			\end{center}
		}
		\vspace{-1mm}
		\noindent
		\begin{tabular}{c}
			\hphantom{aaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaa!} \\
			\hline\hline
		\end{tabular}
		
		\vspace{3mm}
		{\flushleft\it Том ??, № ?, стр. ???--??? (202?) \hspace{\fill}{\rm\small УДК \udcs \hspace{-4mm}}}
		\newline
		{\rm\small \textcolor{blue}{https://doi.org/10.33048/semi.2023.20.???}}\hphantom{aaaaaaaaaaaaaaaaaa\;}{\rm\small MSC\ \ \mscs }
	}%\vbox
}

\begin{document}
\begin{thebibliography}{99}
	
\bibitem{BW71} P. Butzer, U. Westphal,
\href{https:/doi.org/10.1512/iumj.1971.20.20107}{\it The mean ergodic theorem and saturation}, Indiana Univ. Math. J., \textbf{20}:12 (1971), 1163–-1174. Zbl 0217.45202

\bibitem{Ga75} V.F. Gaposhkin,
\href{https://doi.org/10.1070/IM1975v009n06ABEH001522}{Convergence of series connected with stationary sequences}, Mathematics of the USSR-Izvestiya, \textbf{9}:6 (1977), 1297--1321. Zbl 0349.60032

\bibitem{Se14} V.V. Sedalishchev,
\href{https://doi.org/10.1134/S0037446614020165}
{\it Interrelation between the convergence rates in von Neumann's and Birkhoff's ergodic theorems}, Siberian Math. J., \textbf{55}:2 (2014), 336--348. Zbl 1351.37031

\bibitem{CL17} G. Cohen, M. Lin,
\href{https://zbmath.org/1474.47032}{\it Double coboundaries for commuting contractions}, Pure Appl. Funct. Anal.,\textbf{2}:1 (2017), 11--36. Zbl 1474.47032

\bibitem{Brad} R.C. Bradley, 
\href{https://doi.org/10.1017/S1446788708000736}{\it An $L^p$ ‘cousin of coboundary’ theorem for random fields}, J. Aust. Math. Soc., {\bf 85} (2008), 51--58. Zbl 1153.60348

\bibitem{CL21} G. Cohen, M. Lin, 
\href{https://doi.org/10.1512/iumj.2021.70.8547}\it {Joint and double coboundaries of commuting contractions}, 
Indiana Univ. Math. J., {\textbf 70}:4 (2021), 1355--1394. Zbl 07403323

\bibitem{Rob} E.A. Robinson
\href{https://doi.org/10.1090/S0002-9939-1960-0111068-9}{\it Sums of stationary random variables}, Proc. Amer. Math. Soc., \textbf{11}:1 (1960), 77--79. Zbl 0099.12602 

\bibitem{KPH23} A.G. Kachurovskii, I.V. Podvigin, A.J. Khakimbaev,
\href{https://doi.org/10.1134/S0001434623050073}{\it Uniform convergence on subspaces in von Neumann ergodic theorem with discrete time}, Mathematical Notes, \textbf{113}:5 (2023), 680--693.
Zbl 1523.37012

\bibitem{Ka99}
A.G. Kachurovskii, 
\href{https://doi.org/10.1023/A:1012425724804}{\it Convergence of averages in the ergodic theorem for groups $Z^d$}, J. Math. Sci. (New York), \textbf{107}:5 (2001), 4231--4236. Zbl 0987.28014

\bibitem{KPTKh24} A.G. Kachurovskii, I.V. Podvigin, V.E. Todikov, A.J. Khakimbaev,
\href{https://doi.org/10.1134/S0037446624010099}{\it A spectral criterion for power-law convergence rate in the ergodic theorem for $Z^d$ and $R^d$ actions},
Siberian Math. J., \textbf{65}:1 (2024), 76--95. Zbl 1536.37004

\bibitem{KrL} U. Krengel, M. Lin, 
\href{https://doi.org/10.1007/BF01236264}{\it On the range of the generator of a Markovian semigroup}, Math. Zeit., {\bf 185} (1984), 553--565. Zbl 0525.60080 

\bibitem{FLW11} V.P. Fonf, M. Lin, P. Wojtaszcyk, 
\href{https://www.impan.pl/en/publishing-house/journals-and-series/colloquium-mathematicum/all/124/2/87933/poisson-s-equation-and-characterizations-of-reflexivity-of-banach-spaces}{\it Poisson's equation and characterizations of reflexivity of Banach spaces}, Colloquium Math.,  \textbf{124}:2 (2011), 225--235. Zbl 1242.46016

\bibitem{KPT23} A.G. Kachurovskii, I.V. Podvigin, V.E. Todikov, 
\href{https://doi.org/10.33048/semi.2023.20.016}{\it Uniform convergence on subspaces in von Neumann’s ergodic theorem with continuous time}, Siberian Electronic Math. Reports, \textbf{20}:1 (2023), 183--206. Zbl 07896765

\bibitem{Ka96}
A. G. Kachurovskii, 
\href{https://doi.org/10.1070/RM1996v051n04ABEH002964}{\it The rate of convergence in ergodic theorems}, Russian Math. Surveys, \textbf{51}:4 (1996), 653--703. Zbl 0880.60024

\bibitem{KaV99} A.M. Vershik, A.G. Kachurovskii,
\href{https://diffjournal.spbu.ru/EN/numbers/1999.1/article.1.2.html}{\it Rates of convergence in ergodic theorems for locally finite groups and reversed martingales}, Differ. Uravn. Protsessy Upr., {\bf 1} (1999), 19--26. (Russian) Zbl 1506.60013

\bibitem{Po24Spb} I.V. Podvigin,
\href{https://www.mathnet.ru/eng/aa1930}{\it A criterion for the power-law rate of convergence of ergodic means for unitary actions of $\mathbb{Z}^d$	and $\mathbb{R}^d$}, Algebra i Analiz, \textbf{36}:4 (2024), 148--164. (Russian)

\bibitem{Po20} I.V. Podvigin,
\href{https://doi.org/10.33048/semi.2020.17.041}{\it Lower bound of the supremum of ergodic averages for ${\mathbb{Z}^d}$ and ${\mathbb{R}^d}$-actions}, Siberian Electronic Math. Reports, {\bf 17} (2020),  626--636. Zbl 1434.37002

\bibitem{Kr} U. Krengel,
\href{https://doi.org/10.1515/9783110844641}{\it Ergodic theorems} (De Gruyter Studies in Mathematics, V.6), Walter de Gruyter, Berlin--New York, 1985. Zbl 0575.28009

\bibitem{Zas75} T. Zaslavsky, 
\href{https://doi.org/10.1090/memo/0154}{\it Facing up to arrangements: face-count formulas for partitions of space by hyperplane}, Memoirs of AMS, \textbf{1}:154 (1975), AMS, Providence RI. Zbl 0296.50010

\bibitem{F82} N.I. Fel'dman,	
\href{https://search.rsl.ru/ru/record/01001117685}{\it Sed'maya problema Gil'berta}, Izd-vo MGU, Moscow, 1982. (Russian) Zbl 0515.10031

\bibitem{MP90} Yu. I. Manin, A.A. Panchishkin, 
\href{https://www.mathnet.ru/eng/intf147}{\it Introduction to number theory}, Number theory – 1, Itogi Nauki i Tekhniki. Ser. Sovrem. Probl. Mat. Fund. Napr., {\bf 49} (1990),  5--341, VINITI, Moscow. (Russian) Zbl 0732.11002

\bibitem{KM53} P.G. Kontorovich, D.I. Mil'man, 
\href{https://www.mathnet.ru/eng/rm8163}{\it On a method of N.I. Lobachevskii for finding integer solutions of linear homogeneous equations with integer coefficients}, Uspekhi Mat. Nauk, \textbf{8}:1(53) (1953), 145--149. (Russian) Zbl 0050.03610

\bibitem{EW11} M. Einsiedler, T. Ward, 
\href{https://doi.org/10.1007/978-0-85729-021-2}{\it Ergodic theory with a view towards number theory}, (Graduate Texts in Mathematics V. 259), London, Springer, 2011. Zbl 1206.37001

\bibitem{N51} I. Niven, 
\href{https://doi.org/10.1090/S0002-9904-1951-09543-9}{\it The asymptotic density of sequences}, Bull. Amer. Math. Soc. \textbf{57}:6 (1951), 420--434. Zbl 0044.03603

\bibitem{Bo} V.I. Bogachev, 
\href{https://doi.org/10.1007/978-3-540-34514-5}{\it Measure theory}, Vol.~2, Berlin, Springer, 2007. Zbl 1120.28001  


	
	
\end{thebibliography}

	
\renewcommand{\refname}{References}
\thispagestyle{empty}

\title[\sc Maximum rate of norm convergence]{\large Maximum rate of norm convergence \\ in the ergodic theorem \\ for groups $\mathbb{Z}^d$ and $\mathbb{R}^d$}

\author[A.G. KACHUROVSKII, I.V. PODVIGIN, A.J. KHAKIMBAEV]{
	{\bf \large \textsc{A.G. KACHUROVSKII}}\protect\orcidicon{0000-0002-2747-2660}{\bf ,} 
	{\bf \large \textsc{I.V. PODVIGIN,}} \\  
	{\bf \large \textsc{A.J. KHAKIMBAEV}}
	\\ \\
	\textit{Представлено} \textsc{???????}
	%\textit{(заполняется редактором)}
}%
\address{Alexander Grigorievich  Kachurovskii  % Контактные данные всех авторов и место работы указываются только на английском языке
	\newline\hphantom{iii} Sobolev Institute of Mathematics,
	\newline\hphantom{iii} pr. Koptyuga, 4,
	\newline\hphantom{iii} 630090, Novosibirsk, Russia}%
\email{\textcolor{blue}{agk@math.nsc.ru}}%

\address{Ivan Viktorovich Podvigin  % Контактные данные всех авторов и место работы указываются только на английском языке
	\newline\hphantom{iii} Sobolev Institute of Mathematics,
	\newline\hphantom{iii} pr. Koptyuga, 4,
	\newline\hphantom{iii} 630090, Novosibirsk, Russia}%
\email{\textcolor{blue}{ipodvigin@math.nsc.ru}}%

\address{Aziz Jamalatdin uli Khakimbaev  % Контактные данные всех авторов и место работы указываются только на английском языке
	\newline\hphantom{iii} Novosibirsk State University,
	\newline\hphantom{iii} st. Pirogova, 1,
	\newline\hphantom{iii} 630090, Novosibirsk, Russia}%
\email{\textcolor{blue}{a.khakimbaev@g.nsu.ru}}%

\thanks{\sc Kachurovskii A.G., Podvigin I.V., Khakimbaev A.J., Maximum rate of norm convergence in the ergodic theorem for groups $\mathbb{Z}^d$ and $\mathbb{R}^d$}
\thanks{\copyright \ 2025 Kachurovskii A.G., Podvigin I.V., Khakimbaev A.J.}
\thanks{\rm The work was carried out in the framework of the State Task to the Sobolev Institute of
	Mathematics (Project FWNF-2022-0004).}
\thanks{\it Received ???? 2024, published ???? 2025}%


\semrtop \vspace{1cm}

\maketitle 

{\small
	\begin{quote}
		\noindent{\bf Abstract:} For $d$ pairwise commuting automorphisms (flows) of a probability space, ergodic averages over parallelepipeds are considered. It is shown that the maximum rate of their convergence in the $L_p$-norm is ${\mathcal{O}(\frac{1}{t_1t_2\cdots t_d}) }.$ A spectral criterion is also obtained for the maximum convergence rate in the $L_2$-norm.\medskip
		
		\noindent{\bf Keywords:} rates of convergence in ergodic theorems, spectral measure, coboundaries, bundle of hyperplanes.
	\end{quote}
}

\bigskip	
	
\section{Introduction}

{\bf\thesection.1.} Let $T$ be a measure-preserving transformation of a probability measure space  $(\Omega,\mu).$ It is well known that ergodic averages 
$$
A^T_nf(\omega)=\frac{1}{n}\sum_{k=0}^{n-1}f(T^k\omega),\ \ f\in L_p(\Omega,\mu), \ p\in[1,+\infty)
$$	
converge in norm as ${n\to\infty}$ to their limit $f^*$ at a rate no faster than ${\mathcal{O}(\frac{1}{n})}.$ Namely, if ${\|A^T_nf-f^*\|_p=o(1/n)}$ as ${n\to\infty},$ then ${f=f^*}.$ For ${p>1}$ this result was obtained by Butzer and Westphal~\cite[Theorem~1]{BW71}, where a more general case of power bounded operators  in a reflexive Banach space was considered. Gaposhkin obtained a new proof for the  $L_2$-convergence case in~\cite[Corollary~5]{Ga75}, where the spectral representation of stationary sequences was used. In this case, the fact about the maximum rate immediately followed from the inequality
$$
\overline{\lim\limits_{n\to\infty}}n\|A_n^Tf\|_2>0
$$
for a non-zero function~$f.$ Sedalishchev in~\cite[Lemma~1]{Se14} obtained a simple proof of this inequality for all ${L_p,\ p\in[1,+\infty]},$ strengthening it to the following form:
$$
\overline{\lim\limits_{n\to\infty}}n\|A_n^Tf\|_p\geq\frac{\|f\|_p}{2}.
$$

The purpose of this note is to obtain an analogue of the maximum rate of convergence statement for ergodic averages
$$
A_{\vec{n}}f(\omega)=A^{T_1}_{n_1}\cdots A^{T_d}_{n_d}f(\omega),\ \ \vec{n}=(n_1,...,n_d)\in\mathbb{N}^d,
$$
where $T_k,\ k=1,...,d,$ are pairwise commuting automorphisms of the probability measure space~${(\Omega,\mu)};$ and also for ergodic averages with continuous time
$$
\mathcal{A}_{\vec{t}}f(\omega)=\frac{1}{t_1t_2\cdots t_d}\int_{[0,\vec{t}]}f(T_1^{u_1}\cdots T_d^{u_d}\omega)\,d\vec{u}, \ \ \vec{t}=(t_1,...,t_d)\in\mathbb{R}^d,
$$
where $\{T^t_k\}_{t\in\mathbb{R}},\ k=1,...,d,$ are pairwise commuting flows of the probability measure space~${(\Omega,\mu)}.$

We show that for all ${p\in[1,+\infty]}$ the maximum rate of convergence in the $L_p$-norm is ${\mathcal{O}(\frac{1}{n_1n_2\cdots n_d})}$ as ${n_1,...,n_d\to+\infty}$ for discrete time, and ${\mathcal{O}(\frac{1}{t_1t_2\cdots t_d})}$ as ${t_1,...,t_d\to+\infty}$ for continuous time. This is a consequence of Theorem~\ref{Prop1}. Put
$$
\pi(\vec{t})=t_1t_2\cdots t_d,\ \ \vec{t}\in\mathbb{R}^d,
$$
and we will write ${\vec{t}\geq\vec{s}}$ if only ${t_j\geq s_j}$
for each coordinate.

\begin{theorem}\label{Prop1}
 Let ${f\in L_p(\Omega,\mu)},$ ${p\in[1,\infty]};$ then
	$$
	\sup\limits_{\vec{m}\geq\vec{n}}\|A_{\vec{m}}f\|_p\geq\frac{\|f\|_p}{2^{2d}\pi(\vec{n})}\ \ 
	\text{for all}\ \ \vec{n}\in\mathbb{N}^d,
	$$
	$$
	\sup\limits_{\vec{s}\geq\vec{t}}\|\mathcal{A}_{\vec{s}}f\|_p\geq\frac{\sup\limits_{\vec{q}\in E_{\vec{t}}}\left\|\int_{[0,\vec{q}]}f(T_1^{u_1}\cdots T_d^{u_d}\omega)\,d\vec{u}\ \right\|_p}{2^{2d}\pi(\vec{t})}\ \ 
	\text{for all}\ \ \vec{t}\in\mathbb{R}_+^d,
	$$ 
	where ${E_{\vec{t}}=\left\{\vec{q}=(\frac{t_1}{n_1},...,\frac{t_d}{n_d}):\ \vec{n}\in\mathbb{N}^d,\ \vec{n}\geq\vec{t}\right\}\subset(0,1]^d}.$
\end{theorem}

{\bf\thesection.2.} The maximum rate of convergence $\mathcal{O}(\frac{1}{n_1\cdots n_d})$ or $\mathcal{O}(\frac{1}{t_1\cdots t_d})$ of ergodic averages is equivalent to the condition
$$
\sup\limits_{\vec{n}\in\mathbb{N}^d}\left\|\sum_{k_1=0}^{n_1-1}\cdots
\sum_{k_d=0}^{n_d-1}f(T^{k_1}_1\cdots T^{k_d}_d\omega)\right\|_p<\infty\ \ \text{for discrete time},
$$
$$
\sup\limits_{\vec{t}\in\mathbb{R}_+^d}\left\|\int_{[0,\vec{t}]}f(T^{u_1}_1\cdots T^{u_d}_d\omega)\,d\vec{u}\right\|_p<\infty\ \ \text{for continuous time}.
$$
For the case of a single measure-preserving transformation, it is well known that the last inequality holds only for $L_p$-coboundaries (or functions from $L_p$ cohomological to zero), i.e., ${f\in (I-T)L_p(\Omega,\mu)}.$ An excellent exposition of the history of the question can be found in the introduction to Cohen and Lin's paper~\cite{CL17}. In general, it follows from Bradley's paper~\cite{Brad} that $f$ is a $d$-multiple $L_p$-coboundary for all $d$ measure-preserving transformations, i.e.,
$$
f\in(I-T_1)\cdots(I-T_d)L_p(\Omega,\mu).
$$
Cohen and Lin obtained~\cite{CL17},~\cite[Proposition~3.1]{CL21} a spectral criterion for two-fold $L_2$-coboundaries in the case of two commuting transformations. We give this assertion (the equivalence of conditions (1) and (2) below) in the general case ${d\geq1}.$ Note that for a single measure-preserving transformation the spectral criterion is due to Robinson~\cite{Rob}. In a recent note~\cite{KPH23} a similar criterion for uniform convergence on subspaces was obtained.

\begin{theorem}\label{Theorem1}
	Let ${T_1,...,T_d}$ be pairwise commuting automorphisms of the probability measure space ${(\Omega,\mu)}$. Then for $f\in L_2(\Omega,\mu)$ the following conditions are equivalent:
	$$	
	f \in (I-T_1)\cdots(I-T_d)L_2(\Omega,\mu);\eqno(1)
	$$
	$$ \int\limits_{(-\pi,\pi]^d} \frac{d\sigma_{f}(x_1,\dots, x_d)}{\sin ^2 {x_1 \over 2}\cdots \sin ^2 {x_d \over 2}}<\infty;\eqno(2)	
	$$
	$$
	\left\|\frac{1}{n^d}\sum_{k_1 = 0}^{n-1}\dotsb\sum_{k_d = 0}^{n-1}f(T_1^{k_1}\cdots T_d^{k_d}\omega)\right\|_2=O \left({1 \over n^d}\right)\ \ \text{as}\ \ n\to\infty;\eqno(3)
	$$
	$$
	\left\|\frac{1}{\pi(\vec{n})}\sum_{k_1 = 0}^{n_1-1}\dotsb\sum_{k_d = 0}^{n_d-1}f(T_1^{k_1}\cdots T_d^{k_d}\omega)\right\|_2=O\left({1 \over \pi(\vec{n})}\right)\ \ \text{as}\  n_1,...,n_d\to\infty.\eqno(4)
	$$
\end{theorem}

Here ${\sigma_f(x_1,...,x_d)}$ is the spectral measure constructed from the function $f$ and the automorphisms ${T_1,...,T_d}$ (see, for example,~\cite{Ka99, KPTKh24}). In particular, using the spectral measure one can express the norms of ergodic means, namely
$$
\|A_{\vec{n}}f\|_2^2=\frac{1}{n_1^2 \cdots n_d^2}\int\limits_{(-\pi,\pi]^d} \frac{\sin^2\frac{n_1x_1}{2}\cdots \sin^2\frac{n_dx_d}{2} }{\sin^2\frac{x_1}{2}\cdots \sin^2\frac{x_d}{2}}
\, d\sigma_{f}(x_1,\dots,x_d).
$$

A similar result holds for continuous time. To formulate it, we define an analogue of $L_p$-coboundaries. Let $R_j$ be the generator of the group ${\{T^t_j\}_{t\in\mathbb{R}}},$ ${j=1,...,d},$ i.e., in the strong operator topology ${R_j=\lim\limits_{t\to0+}\frac{1}{t}(I-T_j^t)}.$ It is easy to verify that on ${{\mathrm{Dom}}\,R=\cap_{j=1}^d{\mathrm{Dom}}\,R_j}$ the generators commute with each other, so on this domain the operator ${R=R_1\cdots R_d}$ is well defined. We will say that $f$ is a $d$-multiple $L_p$-coboundary if it lies in the image of the operator~$R,$ i.e., there exists a function ${g\in L_p(\Omega,\mu)}$ such that ${f=Rg}.$

\begin{theorem}\label{Theorem2}
	Let ${T^{t_1}_1,...,T^{t_d}_d}$ be pairwise commuting flows of the probability measure space ${(\Omega,\mu)}$. Then for $f\in L_2(\Omega,\mu)$ the following conditions are equivalent:
	$$	
	f \in RL_2(\Omega, \mu);\eqno(1')
	$$
	$$ \int\limits_{\mathbb{R}^d} \frac{d\sigma_{f}(x_1,\dots, x_d)}{x^2_1\cdots x^2_d}<\infty;\eqno(2')	
	$$
	$$
	\left\|\frac{1}{t^d}\int_{[0,t]^d}f(T_1^{u_1}\cdots T_d^{u_d}\omega)\,d\vec{u}\right\|_2=O \left({1 \over t^d}\right)\ \ \text{as}\ \ t\to\infty;\eqno(3')
	$$
	$$
	\left\|\frac{1}{\pi(\vec{t})}\int_{[0,\vec{t}]}f(T_1^{u_1}\cdots T_d^{u_d}\omega)\,d\vec{u}\right\|_2=O\left({1 \over \pi(\vec{t})}\right)\ \ \text{as}\  t_1,...,t_d\to\infty.\eqno(4')
	$$
\end{theorem}

Note that in the case ${d=1}$ the equivalence of $(1')$ and $(4')$ was proved in~\cite{KrL} (see also~\cite{FLW11}), and the equivalence of $(2')$ and $(3')$ was considered in~\cite{KPT23}.

The problem of transferring known results on convergence rates in
ergodic theorems~\cite{Ka96} from the action of the group $\mathbb{Z}$ to the groups $\mathbb{Z}^d$ and $\mathbb{R}^d$ (and further to the widest possible class of groups) was set to the first co-author of this paper by A.M.~Vershik and A.M.~Stepin in the mid-1990s; its solution was started at the same time in~\cite{Ka99} and ~\cite{KaV99}. Theorems~\ref{Theorem1} and~\ref{Theorem2}, together with the results of~\cite{Po24Spb}, complete the solution of this problem for power rates of convergence in the norm (with all possible exponents) for the actions of the groups
$\mathbb{Z}^d$ and $\mathbb{R}^d$.

\section{Proof of Theorem~\ref{Prop1}}

{\bf\thesection.1.} Consider a mapping ${\mathcal{L}:\mathbb{N}^d\to\mathbb{N}^d}$ such that
$$ {\vec{n}<\mathcal{L}(\vec{n})}\ \ \text{for all}\ \ {\vec{n}\in\mathbb{N}^d},\ \text{and}\ \
{\mathcal{L}(\vec{n})\leq \mathcal{L}(\vec{m})}\ \text{for}\ \ {\vec{n}\leq \vec{m}}.
$$
We will say that a net $\{x_{\vec{n}}\}_{\vec{n}\in\mathbb{N}^d}$ of elements of some normed space ${(X, \|\cdot\|)}$ is $(a,\mathcal{L})$-recurrent
for some~${a>0}$ if for any~${\vec{n}\in\mathbb{N}^d}$ there exists
an element~${s(\vec{n})\in\mathbb{N}^d}$ such that ${\vec{n}\leq s(\vec{n})<
	\mathcal{L}(\vec{n})}$ and ${\|x_{s(\vec{n})}\|\geq a}.$ The following theorem holds.

\begin{theorem}\label{Theorem3}
	Let $\{x_{\vec{n}}\}_{\vec{n}\in\mathbb{N}^d}$
	be a ${(a, \mathcal{L})}$-recurrent net; then for
	all~${\vec{n}\in\mathbb{N}^d}$
	
	$$
	\sup_{\vec{m}\geq
		\vec{n}}\left\|\frac{1}{\pi(\vec{m})}\sum_{0\leq\vec{k}<\vec{m}}x_{\vec{k}}\right\|
		\geq\frac{a}{2^{d}\pi(\mathcal{L}(\vec{n}))}.
	$$
\end{theorem}

Theorem~\ref{Theorem3} is proved in~\cite{Po20} for complex-valued nets. The proof of the general case for nets with values in a normed space is word for word the same as the original case, since essentially only the triangle inequality for the norm is used.

Theorem~\ref{Theorem3} immediately implies Theorem~\ref{Prop1} for the group time~${\mathbb{Z}^d}$. Indeed, for the function ${f\in L_p(\Omega,\mu)}$, consider the net ${x_{\vec{n}}=f(T_1^{n_1}\cdots T_d^{n_d}\omega)}$ and ${\mathcal{L}(\vec{n})=\vec{n}+(1,...,1)}.$ Taking ${a=\|f\|_p},$ we obtain that the net is ${(a,\mathcal{L})}$-recurrent: ${s(\vec{n})=\vec{n}}$ and ${\|x_{s(\vec{n})}\|_p=a}.$
Therefore, the inequality from Theorem~\ref{Theorem3} is valid, which after applying the estimate ${\pi(\vec{n}+(1,...,1))\leq2^d\pi(\vec{n})}$ becomes exactly the inequality from Theorem~\ref{Prop1}.

Now let's consider the continuous case, i.e., the group time ${\mathbb{R}^d},$ and use the already proven inequality for discrete time. Fix $\vec{t}$ and take an arbitrary vector ${\vec{q}\in E_{\vec{t}}}.$ Put
$$
g_{\vec{q}}(\omega)=\int_{[0,\vec{q}]}f(T_1^{u_1}\cdots T_d^{u_d}\omega)\,d\vec{u},\ \ S_j=T_j^{q_j},\ \ j=1,...,d.
$$
Let us consider the partition of the interval ${[0,\vec{n}]}$ into unit cubes, i.e.,
$$
[0,\vec{n}]=\bigcup_{0\leq\vec{k}\leq\vec{n}}P_{\vec{k}},\ {P_{\vec{k}}=[0,1]^d+\vec{k}},\ \ \ {\vec{k}\leq\vec{n}}.
$$ 
We denote the coordinate-wise multiplication of vectors in~${\mathbb{R}^d}$ as
$${\vec{t}\odot\vec{s}=(t_1s_1,...,t_ds_d)}.$$
It is easy to see that
\begin{multline*}
\sum_{0\leq\vec{k}\leq\vec{n}}\int_{P_{\vec{k}}} f(S_1^{u_1}\cdots S_d^{u_d}\omega)\,d\vec{u}=\sum_{0\leq\vec{k}\leq\vec{n}}\int_{[0,1]^d}f(S_1^{u_1+k_1}\cdots S_d^{u_d+k_d}\omega)\,d\vec{u}=\\
=\frac{1}{\pi(\vec{q})}\sum_{0\leq\vec{k}\leq\vec{n}}\int_{[0,\vec{q}]}f(T_1^{t_1}S_1^{k_1}\cdots T_d^{t_d}S_d^{k_d}\omega)\,d\vec{t}=\frac{\pi(\vec{n})}{\pi(\vec{q})}A_{n_1}^{S_1}\cdots A_{n_d}^{S_d}g_{\vec{q}}(\omega).
\end{multline*}
From here, taking into account that ${\vec{t}=\vec{n_0}\odot\vec{q}}$ for some ${\vec{n_0}\geq\vec{t}},$ we obtain

\begin{multline*}
\sup\limits_{\vec{s}\geq\vec{t}}\|\mathcal{A}_{\vec{s}}f\|_p\geq\sup\limits_{\vec{s}\geq\vec{t}, \vec{s}=\vec{n}\odot\vec{q}, \vec{n}\in\mathbb{N}^d}\|\mathcal{A}_{\vec{s}}f\|_p=\\
=\sup\limits_{\vec{n}\geq\vec{n_0}}\frac{1}{\pi(\vec{n}\odot\vec{q})}\left\|\int_{[0,\vec{n}\odot\vec{q}]}f(T_1^{u_1}\cdots T_d^{u_d}\omega)\,d\vec{u}\right\|_p=\\
=\sup\limits_{\vec{n}\geq\vec{n_0}}\frac{1}{\pi(\vec{n})}\left\|\int_{[0,\vec{n}]}f(T_1^{q_1u_1}\cdots T_d^{q_du_d}\omega)\,d\vec{u}\right\|_p=\\
=\sup\limits_{\vec{n}\geq\vec{n_0}}\frac{1}{\pi(\vec{n})}\left\|\sum_{0\leq\vec{k}\leq\vec{n}}\int_{P_{\vec{k}}} f(S_1^{u_1}\cdots S_d^{u_d}\omega)\,d\vec{u}\right\|_p=\\
=\sup\limits_{\vec{n}\geq\vec{n_0}}\frac{1}{\pi(\vec{q})}\left\|A_{n_1}^{S_1}\cdots A_{n_d}^{S_d}g_{\vec{q}}(\omega)\right\|_p\geq\frac{\|g_{\vec{q}}\|_p}{2^{2d}\pi(\vec{q})\pi(\vec{n_0})}=\frac{\|g_{\vec{q}}\|_p}{2^{2d}\pi(\vec{t})}.
\end{multline*}
Since ${\vec{q}}\in E_{\vec{t}}$ was arbitrary, we can take the supremum over all such vectors. Theorem~\ref{Prop1} is completely proved.

{\bf Remark.} V.V. Ryzhikov proposed the following argument to justify the inequality for the discrete case in Theorem~\ref{Prop1}. The following statement is true: for any ${\vec{n}\in\mathbb{N}^d}$ there exists ${\vec{n'}\geq\vec{n}},$ such that ${\|\vec{n}-\vec{n'}\|_\infty\leq1}$ and
$$
\left\|\sum_{k_1=0}^{n'_1-1}\cdots\sum_{k_d=0}^{n'_d-1}f(T^{k_1}_1\cdots T^{k_d}_d\omega)\right\|_p\geq\frac{\|f\|_p}{2^d}.
$$
This inequality strengthens Sedalishchev's inequality from the one-dimensional case.
The proof is by induction. We will only show the basis for ${d=1}.$ From the equality
$$
\sum_{k=0}^nf(T^k\omega)-\sum_{k=0}^{n-1}f(T^k\omega)=f(T^n\omega)
$$
it follows that the norms of both sums on the right-hand side cannot be less than ${\|f\|_p/2}$ at the same time.

Now, having the strengthened Sedalishchev's inequality, we obtain
$$
\sup\limits_{\vec{m}\geq\vec{n}}\|A_{\vec{m}}f\|_p\geq\sup\limits_{\|\vec{n'}-\vec{n}\|_\infty\leq1}\|A_{\vec{n'}}f\|_p\geq\frac{\|f\|_p}{2^d\pi(\vec{n'})}\geq\frac{\|f\|_p}{2^{2d}\pi(\vec{n})}.
$$


{\bf\thesection.2.} Let us show how the statement about the maximum rate of convergence of ergodic means follows from Theorem~\ref{Prop1}.

Let ${\|A_{\vec{n}}(f-f^*)\|_p=o\left(\frac{1}{n_1\cdots n_d}\right)}$ for ${n_1,...,n_d\to+\infty}.$ Then
$$
\frac{\|f-f^*\|_p}{2^{2d}\pi(\vec{n})}\leq\sup\limits_{\vec{m}\geq\vec{n}}\|A_{\vec{m}}(f-f^*)\|_p=o\left(\frac{1}{\pi(\vec{n})}\right),
$$
whence ${\|f-f^*\|_p=o(1)},$ i.e., ${f(\omega)=f^*(\omega)}$ a.e.	

For continuous time, also assume that ${\|\mathcal{A}_{\vec{t}}(f-f^*)\|_p=o\left(\frac{1}{t_1\cdots t_d}\right)}$ as ${t_1,...,t_d\to+\infty}.$ Then
$$
\frac{\sup_{\vec{q}\in E_{\vec{t}}}\|\int_{[0,\vec{q}]}f(T_1^{u_1}\cdots T_d^{u_d}\omega)\,d\vec{u}-\pi(\vec{q})f^*\|_p}{2^{2d}\pi(\vec{t})}\leq\sup\limits_{\vec{s}\geq\vec{t}}\|\mathcal{A}_{\vec{s}}(f-f^*)\|_p=o\left(\frac{1}{\pi(\vec{t})}\right),
$$
whence ${\sup_{\vec{q}\in E_{\vec{t}}}\|\int_{[0,\vec{q}]}f(T_1^{u_1}\cdots T_d^{u_d}\omega)\,d\vec{u}-\pi(\vec{q})f^*\|_p=o(1)}$ as ${t_1,...,t_d\to+\infty}.$

Let ${\vec{t}=n\vec{t_0},\ n\to\infty},$ and the vector $\vec{t_0}$ be fixed. Since
${E_{\vec{t_0}}\subset E_{n\vec{t_0}}\subset(0,1]^d},$ then
${\sup_{\vec{q}\in E_{\vec{t_0}}}\|\int_{[0,\vec{q}]}f(T_1^{u_1}\cdots T_d^{u_d}\omega)\,d\vec{u}-\pi(\vec{q})f^*\|_p=o(1)}$ as ${n\to\infty},$ therefore, for any ${\vec{q}\in E_{\vec{t}_0}},$ 
$$
\left\|\int_{[0,\vec{q}]}f(T_1^{u_1}\cdots T_d^{u_d}\omega)\,d\vec{u}-\pi(\vec{q})f^*\right\|_p=0.
$$
Then from the local ergodic theorem~\cite[\S7.2]{Kr} we obtain a.e. equality
$$
f^*(\omega)=\lim\limits_{E_{\vec{t_0}}\ni\vec{q}\to0}\frac{1}{\pi(\vec{q})}\int_{[0,\vec{q}]}f(T_1^{u_1}\cdots T_d^{u_d}\omega)\,d\vec{u}=f(\omega).
$$

\section{Functions $\mathcal{J}_d(x)$ and $\mathcal{K}_d(x)$}

{\bf\thesection.1.} Let us consider an auxiliary problem, interesting in itself and useful for proving the spectral criterion for the maximum possible rate of convergence, i.e., for Theorems~\ref{Theorem1} and \ref{Theorem2}. For ${x=(x_1,...,x_d)}$ we define the functions
$$
\mathcal{J}_d(x)=\lim\limits_{T\to\infty}\frac{1}{T}\int_0^T\prod\limits_{k=1}^d\sin^2(x_ks/2)\,ds, \ {x\in\mathbb{R}^d};
$$
$$ \mathcal{K}_d(x)=\lim\limits_{N\to\infty}\frac{1}{N}\sum\limits_{n=1}^N\prod\limits_{k=1}^d\sin^2(x_kn/2), \ x\in\left(-\pi,\pi\right]^d.
$$
Our goal is to find the set of zeros of these functions and to show that their set of values is finite. The reader interested in the proof of Theorems~\ref{Theorem1} and \ref{Theorem2} can skip to the next sections and return to \S3 later.

To calculate the limits, we use the following construction. Consider in the space ${\mathbb{R}^d}$ the bundle~$\mathcal{L}(a)$ of all hyperplanes defined by the equalities ${x_1\varepsilon_1+x_2\varepsilon_2+...+x_d\varepsilon_d}=a,$ where $a\in\mathbb{R}$ is fixed, and ${\varepsilon_k\in\{-1,0,1\}}$ and not all of them are equal to zero simultaneously. It is clear that ${\mathcal{L}(a)=\mathcal{L}(-a)}$ for each ${a\in\mathbb{R}}.$ The index of a hyperplane~$L$ from the bundle ${\mathcal{L}(a)}$ is the number of all nonzero coefficients in the equation defining it, i.e., the quantity ${{\rm ind}L=\sum\limits_{k=1}^d|\varepsilon_k|=\|\varepsilon\|_1}.$ In the case $a=0$, the hyperplanes from the bunch $\mathcal{L}(0),$ defined by the vectors ${\varepsilon}$ and ${-\varepsilon},$ will be considered different. Many different combinatorial problems are associated with such constructions; see, for example,~\cite{Zas75}. We also set
$$
\mathcal{L}_d=\mathcal{L}(0)\cup\mathcal{L}(2\pi)\cup\cdots\cup\mathcal{L}\left(2\pi\left\lfloor d/2\right\rfloor\right).
$$

\begin{proposition}\label{Pr_forMax} The following equalities are true
	$$
	\mathcal{J}_d(x)=\frac{1}{2^d}\left(1+\sum\limits_{k=1}^d\frac{(-1)^k}{2^{k}}\sum\limits_{\stackrel{L\in\mathcal{L}(0)}{{\rm ind}L=k}}I_{\{x\in L\}}\right),\ \ x\in\mathbb{R}^d;
	$$
	$$
	\mathcal{K}_d(x)=\frac{1}{2^d}\left(1+\sum\limits_{k=1}^d\frac{(-1)^k}{2^{k}}\sum\limits_{\stackrel{L\in\mathcal{L}_d}{{\rm ind}L=k}}I_{\{x\in L\}}\right),\ \ x\in(-\pi,\pi]^d.
	$$
\end{proposition}
\begin{proof}
	Let us prove both equalities simultaneously. Let $\nu$ be the Lebesgue measure on $\mathbb{R}$ or the counting measure on ${\mathbb{Z}}.$ Then both limits in the definitions of the functions ${\mathcal{J}_d}$ and ${\mathcal{K}_d}$ are written in the same way.
	We have
	$$
	\lim\limits_{T\to\infty}\frac{1}{T}\int\limits_{(0,T]}\prod\limits_{k=1}^d\sin^2(x_ks/2)\,d\nu(s)=
	\lim\limits_{T\to\infty}\frac{1}{2^dT}\int\limits_{(0,T]}\prod\limits_{k=1}^d(1-\cos(x_ks))\,d\nu(s).
	$$	
	Using the formula for the product of cosines, we rewrite the product
	\begin{multline*}
	\prod\limits_{k=1}^d\left( 1-\cos(x_ks)\right)=1-\sum\limits_{k=1}^d\cos(x_ks)+\sum\limits_{1\leq i<j\leq d}\cos(x_is)\cos(x_js)+...\\
	+(-1)^k\sum\limits_{1\leq i_1<...<i_k\leq d}\cos(x_{i_1}s)\cdots\cos(x_{i_k}s)+...+(-1)^d\cos(x_1s)\cdots\cos(x_ds)=\\
	=1+\sum_{k=1}^d\frac{(-1)^k}{2^{k-1}}\frac{1}{2}\sum_{\|\varepsilon\|_1=k}\cos((x_1\varepsilon_1+x_2\varepsilon_2+...+x_d\varepsilon_d)s).
	\end{multline*}
The summation $\sum\limits_{\|\varepsilon\|_1=k}$ is over all vectors ${\varepsilon=(\varepsilon_1,...,\varepsilon_d)},$ $\varepsilon_k\in\{-1,0,1\}$ with the norm ${\|\varepsilon\|_1=k}.$ Since the vectors $\varepsilon$ and $-\varepsilon$ make the same contribution, each sum has an additional factor $\frac{1}{2}.$
It remains to calculate the limits of the form ${\lim_{T\to\infty}\frac{1}{T}\int_{(0,T]}\cos(ys)\,d\nu(s)},$ where ${y=\sum_{k=1}^dx_k\varepsilon_k.}$

If $\nu$ is the Lebesgue measure, then the limit is $I_{\{y=0\}}.$ Let's check that for the counting measure on $\mathbb{Z}$ there will be a similar result. Using the known formulas for the sum of cosines, we get
\begin{multline*}
\lim\limits_{N\to\infty}\frac{1}{N}\sum_{n=1}^N\cos(ny)=\\
=\lim\limits_{N\to\infty}\frac{1}{N}\left(\frac{\cos(Ny/2)\sin((N+1)y/2)}{\sin(y/2)}-1\right)=\sum\limits_{k=-\infty}^{+\infty} I_{\{y=2\pi k\}}.
\end{multline*}
	The last sum is in fact finite and equals ${\sum\limits_{k=-\lfloor d/2\rfloor}^{\lfloor d/2\rfloor}I_{\{y=2\pi k\}}}$, since for ${x\in\left(-\pi, \pi\right]^d}$ the point ${y\in(-d\pi,d\pi].}$
	Putting all the calculations together, we obtain the required equalities.
	\end{proof}
	It immediately follows from Proposition~\ref{Pr_forMax} that the functions $\mathcal{J}_d$ and $\mathcal{K}_d$ take a finite number of values. In addition, for all ${x\in(-\pi,\pi]^d}$ such that ${x\not\in\mathcal{L}_d\setminus\mathcal{L}(0)},$ the equality ${\mathcal{J}_d(x)=\mathcal{K}_d(x)}$ holds; in particular, for all ${x\in\left(-\frac{\pi}{d},\frac{\pi}{d}\right]^d}.$

{\bf\thesection.2.} Let us find the zeros of the function~${\mathcal{J}_d},$ and estimate its minimum positive value.
\begin{proposition} The following statements hold for the function $\mathcal{J}_d$:
	
	(i) ${\mathcal{J}_d(x)=\frac{1}{2\pi}\int_0^{2\pi}\prod\limits_{k=1}^d\sin^2(x_ks/2)\,ds}$ for all ${x\in\mathbb{Z}^d}.$
	
	(ii) ${\mathcal{J}_d(x)=0}$ if and only if ${\prod\limits_{k=1}^dx_k=0}.$
	Moreover, for the minimal value ${j_d:=\min\limits_{\prod\limits_{k=1}^dx_k\neq0}\mathcal{J}_d(x)}$
	we have the estimate
	$$ 
	\frac{1}{2(2d+1)d^{(d-1)(2d+1)}}\leq j_d\leq\frac{1}{2^d}.
	$$
\end{proposition}

\begin{proof} 
Considering that for ${x\in\mathbb{Z}^d}$ the squares of the sines involved in the product will be $2\pi$-periodic, we obtain
	\begin{multline*}
	\mathcal{J}_d(x)=\lim\limits_{T\to\infty}\frac{1}{T}\int_0^T\prod\limits_{k=1}^d\sin^2(x_ks/2)\,ds=
	\lim\limits_{N\to\infty}\frac{1}{2\pi N}\int_0^{2\pi N}\prod\limits_{k=1}^d\sin^2(x_ks/2)\,ds=\\
	=\lim\limits_{N\to\infty}\frac{1}{2\pi N}\sum_{n=0}^{N-1}\int_{2\pi n}^{2\pi (n+1)}\prod\limits_{k=1}^d\sin^2(x_ks/2)\,ds=\\
	=\lim\limits_{N\to\infty}\frac{1}{2\pi N}\sum_{n=0}^{N-1}\int_{0}^{2\pi}\!\!\prod\limits_{k=1}^d\sin^2(x_k(t+2\pi n)/2)\,dt=\\
	=\lim\limits_{N\to\infty}\frac{1}{2\pi N}\sum_{n=0}^{N-1}\int_{0}^{2\pi}\!\!\prod\limits_{k=1}^d\sin^2(x_kt/2)\,dt
	=\frac{1}{2\pi}\int_{0}^{2\pi}\!\!\prod\limits_{k=1}^d\sin^2(x_kt/2)\,dt.
	\end{multline*}
	
	The function $\mathcal{J}_d$ is a homogeneous function of order zero, i.e., for any ${\alpha\in\mathbb{R}\setminus\{0\}}$ the equality ${\mathcal{J}_d(\alpha x)=\mathcal{J}_d(x)}$ is true. If the point ${x\in\mathbb{R}^d}$ does not lie in any hyperplane of the bunch~$\mathcal{L}(0),$ then ${\mathcal{J}_d(x)=2^{-d}}.$ The function $\mathcal{J}_d$ takes the same values on linear subspaces that are intersections of hyperplanes of the bunch~$\mathcal{L}(0).$ Since any such intersection contains integer points, the values of the function $\mathcal{J}_d$ are completely determined by its values on $\mathbb{Z}^d.$ Moreover, it can be argued that it is sufficient to take integer points from some cube. To find the boundary of this cube, we use the following statement from the theory of Diophantine equations (see~\cite[Theorem~6.1]{F82}, \cite[Chapter~1, \S2.2]{MP90}, and also~\cite{KM53}). Let ${m<d}$, ${a_{i,j}\in\mathbb{Z}}$ and ${A_i=\sum\limits_{j=1}^d|a_{i,j}|, i=1,...,m};$ then the system of linear homogeneous equations
	$$
	a_{i,1}x_1+a_{i,2}x_2+...+a_{i,d}x_d=0, \ \ i=1,...,m
	$$
	has a non-trivial integer solution satisfying the condition
	$$\max\limits_{1\leq k \leq d}|x_k|\leq\sqrt[d-m]{A_1\cdots A_m}.$$
	In our case, all ${A_i\leq d},$ and the boundary of the cube is estimated as ${\max\limits_{1\leq k \leq d}|x_k|\leq d^{d-1}}.$
	
	From ${(i)}$ and the fact that $\mathcal{J}_d$ is completely determined by the values on ${\mathbb{Z}^d},$ it is immediately clear that $\mathcal{J}_d$ vanishes only if one of the coordinates ${x_k=0}.$ Let us now find the lower bound for $j_d;$ the upper bound is obvious. We will use the representation~${(i)}$, where for the point ${x\in\mathbb{Z}^d}, {x_k\neq0}$ we will assume that ${D=\max\limits_{1\leq k\leq d}|x_k|\leq d^{d-1}}.$ We have
\begin{multline*}
	\mathcal{J}_d(x)=\frac{1}{2\pi}\int_0^{2\pi}\prod\limits_{k=1}^d\sin^2(x_ks/2)\,ds=\\=\frac{1}{\pi}\int_0^{\pi}\prod\limits_{k=1}^d\sin^2(x_ks)\,ds\geq\frac{1}{\pi}\int_0^{\pi/(2D)}\prod\limits_{k=1}^d\sin^2(x_ks)\,ds.
\end{multline*}
	For ${s\in\left[0,\frac{\pi}{2D}\right]}$ we have ${x_ks\in\left[0,\frac{\pi}{2}\right]}$ and hence ${\sin^2(x_ks)\geq(\frac{2}{\pi}x_ks)^2}.$ Thus,
\begin{multline*}
\mathcal{J}_d(x)\geq\frac{1}{\pi}\int_0^{\pi/(2D)}\prod\limits_{k=1}^d\frac{4x^2_ks^2}{\pi^2}\,ds=\frac{4^dx^2_1\cdots x^2_d}{\pi^{2d+1}}\int_0^{\pi/(2D)}s^{2d}\,ds=\\
=\frac{4^dx^2_1\cdots x^2_d}{\pi^{2d+1}}\frac{\pi^{2d+1}}{(2d+1)(2D)^{2d+1}}
=\frac{x^2_1\cdots x^2_d}{2(2d+1)D^{2d+1}}\geq\frac{1}{2(2d+1)d^{(d-1)(2d+1)}}.
	\end{multline*}		
\end{proof}

{\bf\thesection.3.} Let us proceed to finding the zeros of the function~${\mathcal{K}_d(x)}.$

\begin{proposition} ${\mathcal{K}_d(x)=0}$ if and only if ${\prod\limits_{k=1}^dx_k=0}.$
\end{proposition}

\begin{proof}
	As can be seen from the definition of the function ${\mathcal{K}_d},$ if at least one coordinate of the point ${x\in(-\pi,\pi]^d}$ is equal to zero, then the function vanishes. Let us show that there are no other zeros.
	Suppose that there is a point ${x\in(-\pi,\pi]^d},$ such that ${\prod\limits_{k=1}^dx_k\neq0},$ and at the same time ${\mathcal{K}_d(x)=0}.$ We obtain a contradiction using the density theory for subsets of the set of natural numbers. And the transition to this theory is realized using the Koopman---von Neumann lemma (see, for example,~\cite[lemma~2.41]{EW11}): for a bounded sequence of non-negative numbers ${\{a_n\}_{n\geq1}}$ one have the equivalence
	$$
	\lim\limits_{N\to\infty}\frac{1}{N}\sum_{n=1}^Na_n=0\ \ \ \ \Leftrightarrow\ \ \ \ a_n\to0\ \ \text{for}\ \ n\to\infty, \ n\notin J
	$$
	for some set $J$ with zero asymptotic density.
	
	Recall~(see, for example,~\cite{N51}) that the asymptotic density~$d(J)$ of some set ${J=\{j_1,j_2,...\}}$ of natural numbers is the limit (if it exists)
	$$
	d(J)=\lim\limits_{n\to\infty}\frac{\#\{k:j_k\leq n\}}{n}=\lim\limits_{n\to\infty}\frac{n}{j_n}.
	$$
	
	Let ${a_n=\prod\limits_{k=1}^d\sin^2(x_kn/2).}$ Since
	$$
	0=\mathcal{K}_d(x)=\lim\limits_{N\to\infty}\frac{1}{N}\sum_{n=1}^Na_n,
	$$
	then there exists a set ${J\subset\mathbb{N}}$ of asymptotic density 1, along which ${a_n\to0}$ (since the complement of a set of density 0 always has density 1). We will show that this cannot happen. To do this, we will find out along which increasing sequences $m_n$ of natural numbers it will be ${\sin(ym_n)\to0}$ as ${m_n\to\infty}$ for a given ${y\in(-\pi/2,\pi/2]\setminus\{0\}}.$
	
	If $y=\pi\frac{a}{b}, a\in\mathbb{Z}, b\in\mathbb{N},$ ${b\geq2},$ then  ${m_n=bM_n},$ ${M_n\in\mathbb{N}},$ starting from some number $n.$ The maximally dense set will be the arithmetic progression ${m_n=bn},$ ${n\in\mathbb{N}};$ its asymptotic density is~$\frac{1}{b}.$
	
	If ${y=\pi\alpha, \alpha\in\mathbb{R}\setminus\mathbb{Q}},$ then $\{m_n\}_{n\geq1}$ has zero asymptotic density. Indeed, let us first show that ${m_{n+1}-m_n\to +\infty}$ as ${n\to\infty}.$ Let this not be so; then there exists a natural number~$c$ such that for an infinite number of numbers we have ${m_{n+1}-m_n=c}.$ For such numbers we obtain
	\begin{multline*}
	0=\lim_n\sin(ym_{n+1})=\lim_n\sin(ym_n+yc)=\\
	=\lim_n\sin(ym_n)\cos(yc)+\lim_n\cos(ym_n)\sin(yc)=\pm\sin(yc)=\pm\sin(\pi c\alpha).
	\end{multline*}
	The sign $\pm$ depends on the limit of ${\cos(ym_n)}$: is it $1$ or $-1$. Here we conclude that ${c=0}$; it cannot be. Using Stolz theorem for calculating the limits of sequences, we obtain
	$$
	d(\{m_n\})=\lim\limits_{n\to\infty}\frac{n}{m_n}=\lim\limits_{n\to\infty}\frac{n+1-n}{m_{n+1}-m_n}=0.
	$$
	
	Let us return to the sequence ${a_n=\prod\limits_{k=1}^d\sin^2(x_kn/2).}$ The set $J$ along which it tends to zero is the union of the sets $J_k$ along each of which ${\sin(x_kn/2)}$ tends to zero. Among the $J_k$ there are either sets of density zero or arithmetic progressions of the form $\{b_kn\}_{n\geq1}, b_k\in\mathbb{N}, {b_k\geq2}.$ A finite union of sets with zero asymptotic density will be a set with zero density. And a finite union of the arithmetic progressions considered cannot have density 1, since there will always be another arithmetic progression in its complement. Indeed, the arithmetic progression ${\{1+b_1\cdots b_\ell n\}_{n\geq1}}$ has no common terms with any of the progressions ${\{b_kn\}_{n\geq1}}.$ Since otherwise the Diophantine equation
	$$
	1+b_1\cdots b_\ell n=b_km
	$$
	would have a solution ${n,m\in\mathbb{N}};$ it cannot be.
	
	Thus, the asymptotic density of the set $J$ cannot be equal to~1.
\end{proof}

\section{Proof of Theorem~\ref{Theorem1}}

Recall that the equivalence of (1) and (4) follows from~\cite{Brad}. For the remaining items, we prove the following chain: $(4)\Rightarrow(3)\Rightarrow(2)\Rightarrow(4).$

From (4), we can easily obtain (3) by setting ${n_1=\dots =n_d=n}$.

We will show the implication (3)$\Rightarrow$ (2). Let for some constant ${B>0}$ for all natural
${n}$ we have ${\|A_{n}f\|_2^2\leq Bn^{-2d}}.$ Taking into account the representation for the $L_2$-norm of ergodic means, we rewrite this estimate as
$$
B\geq\int\limits_{(-\pi,\pi]^d}\frac{\prod_{i=1}^{d}\sin^2\frac{nx_i}{2}}
{\prod_{i=1}^{d}\sin^2\frac{x_i}{2}}\, d\sigma_{f}(x_1,\dots,x_d).
$$
Summing this inequality over $n$ from 1 to $N$, we obtain
$$
B\geq {1 \over N}\sum \limits_{n=1}^{N}\int\limits_{\left(\pi,\pi\right]^d}\frac{\prod_{i=1}^{d}\sin^2\frac{nx_i}{2}}
{\prod_{i=1}^{d}\sin^2\frac{x_i}{2}}\, d\sigma_{f}(x_1,\dots,x_d).
$$
The Fatou lemma, when passing to the lower limit as ${N\to\infty}$, yields the following inequality:
\begin{multline*}
\liminf_{N\to \infty} {1 \over N}\sum \limits_{n=1}^{N}\int\limits_{\left(-\pi,\pi\right]^d}\frac{\prod_{i=1}^{d}\sin^2\frac{nx_i}{2}}	{\prod_{i=1}^{d}\sin^2\frac{x_i}{2}}\, d\sigma_{f}(x_1,\dots,x_d)\geq\\
\geq \int\limits_{\left(-\pi,\pi\right]^d}\frac{\lim_{N\to \infty}{1 \over N}\sum \limits_{n=1}^{N} \prod_{i=1}^{d}\sin^2\frac{nx_i}{2}}
{\prod_{i=1}^{d}\sin^2\frac{x_i}{2}}\, d\sigma_{f}(x_1,\dots,x_d).
\end{multline*}
Recalling the definition of the function ${\mathcal{K}_d(x)},$ we obtain an estimate close to the required one:
$$
\int\limits_{\left(-\pi,\pi\right]^d}\frac{\mathcal{K}_d(x)}
{\prod_{i=1}^{d}\sin^2\frac{x_i}{2}}\, d\sigma_{f}(x_1,\dots,x_d)\leq B.
$$
From condition (3), applying the conditional expectation operator $\mathbb{E}_j$ with respect to the $\sigma$-algebra of $T_j$-invariant sets, we obtain, on the one hand,
$$
\left\|\mathbb{E}_j\left(\sum_{k_1 = 0}^{n-1}\dotsb\sum_{k_d = 0}^{n-1}f(T_1^{k_1}\cdots T_d^{k_d}x)\right)\right\|_2\leq\left\|\sum_{k_1 = 0}^{n-1}\dotsb\sum_{k_d = 0}^{n-1}f(T_1^{k_1}\cdots T_d^{k_d}x)\right\|_2=\mathcal{O}(1),
$$
and on the other hand, because ${\mathbb{E}_jT_j=\mathbb{E}_j},$
$$
\left\|\mathbb{E}_j\left(\sum_{k_1 = 0}^{n-1}\dotsb\sum_{k_d = 0}^{n-1}f(T_1^{k_1}\cdots T_d^{k_d}x)\right)\right\|_2=n\left\|\sum_{k_1 = 0}^{n-1}\dotsb\sum_{k_d = 0}^{n-1}\mathbb{E}_jf(T_1^{k_1}\cdots \widehat{T^{k_j}_j}\cdots T_d^{k_d}x)\right\|_2.
$$
Putting it all together, we find for the ergodic averages generated by $d-1$ automorphisms (all except $T_j$), the estimate
$$
\|A^{T_1}_{n}\cdots A^{T_d}_n\mathbb{E}_jf\|_2=o\left(\frac{1}{n^{d-1}}\right).
$$
Taking into account Theorem~\ref{Prop1}, we conclude that ${\mathbb{E}_jf=0}$ for each ${j=1,...,d}.$ From this we deduce~\cite[Corollary~1]{KPTKh24}, that the spectral measure ${\sigma_f(\mathbb{O})=0}$ for
$$
\mathbb{O}=\left\{x\in(-\pi,\pi]^d:\ \prod_{i=1}^{d}x_i=0\right\}.
$$ 
By Proposition~3, the function ${\mathcal{K}_d(x)}$ also vanishes on $\mathbb{O}$, therefore  
$$
\int\limits_{\left(-\pi,\pi\right]^d}\frac{\mathcal{K}_d(x)}{\prod_{i=1}^{d}\sin^2\frac{x_i}{2}}\, d\sigma_{f}(x_1,\dots,x_d)\geq \min\limits_{x\notin\mathbb{O}} \mathcal{K}_d(x) \int\limits_{\left(-\pi,\pi\right]^d}\frac{ d\sigma_{f}(x_1,\dots,x_d)}
{\prod_{i=1}^{d}\sin^2\frac{x_i}{2}}.
$$	
Whence, assuming ${\kappa_d=\min\limits_{x\notin\mathbb{O}} \mathcal{K}_d(x)}$ and knowing from Propositions~1 and 3 that ${\kappa_d>0},$ we obtain
$$
\int \limits_{\left(\pi,\pi\right]^d}\frac{d\sigma_{f}(x_1,\dots,x_d)}{\prod_{i=1}^d\sin ^2 {x_i \over 2}}\leq \frac{B}{\kappa_d}<\infty.
$$
We now show the implication (2)$\Rightarrow$(4). If	
$$
\int\limits_{(-\pi,\pi]^d}\frac{d\sigma_{f}(x_1,\dots,x_d)}{\prod_{i=1}^d\sin ^2 {x_i \over 2}}= A<\infty,
$$ 
then for all ${\vec{n}\in\mathbb{N}^d}$
\begin{multline*}
\|A_{\vec{n}}f\|_2^2=\frac{1}{n_1^2 \cdots n_d^2}\int\limits_{(-\pi,\pi]^d} \frac{\sin^2\frac{n_1x_1}{2}\cdots \sin^2\frac{n_dx_d}{2} }{\sin^2\frac{x_1}{2}\cdots \sin^2\frac{x_d}{2}}
\, d\sigma_{f}(x_1,\dots,x_d)\leq\\
\leq\frac{1}{n_1^2 \cdots n_d^2}\int\limits_{(-\pi,\pi]^d} \frac{d\sigma_{f}(x_1,\dots,x_d) }{\sin^2\frac{x_1}{2}\cdots \sin^2\frac{x_d}{2}} \leq A\frac{1}{n_1^2 \cdots n_d^2}.
\end{multline*}
Theorem~\ref{Theorem1} is completely proved.

\section{Proof of Theorem~\ref{Theorem2}}

The equivalence of conditions $(2')$, $(3')$, and $(4')$ is proved word for word, just as in Theorem~\ref{Theorem1} the equivalence of conditions (2), (3), and (4) is proved. The difference is the use of the function ${\mathcal{J}_d}$ (instead of the function ${\mathcal{K}_d}$), whose zeros form the set $\mathbb{O}'=\left\{x\in\mathbb{R}^d:\ \prod_{i=1}^{d}x_i=0\right\}.$
We prove the equivalence of $(1')$ and $(4').$

\begin{proposition}
Let ${T^{t_1},...,T^{t_d}}$ be pairwise commuting flows of a probability measure space  ${(\Omega,\mu)}$. For any function ${f\in L_p(\Omega,\mu)},\ {p\in[1,+\infty)}$ the equivalence holds
	$$
	f\in RL_p(\Omega,\mu)\ \ \ \Leftrightarrow \ \ \
	\sup\limits_{\vec{t}\in\mathbb{R}^d_+}\left\|\int_{[0,\vec{t}]}f(T_1^{u_1} \cdots T_d^{u_d} \omega)\,d\vec{u}\ \right\|_p<\infty. 
	$$
\end{proposition}

\begin{proof}
	It is enough to prove the implication $\Leftarrow.$ Let us reduce the problem to the already known criterion for discrete time. Let
	$$
	\sup\limits_{\vec{t}\in\mathbb{R}^d_+}\left\|\int_{[0,\vec{t}]}f(T_1^{u_1} \cdots T_d^{u_d} \omega)\,d\vec{u}\ \right\|_p=C<\infty.
	$$
	Fix an arbitrary vector ${\vec{q}\in\mathbb{R}_+^d},$ and set, as in the proof of Theorem~1,
	$$
	g_{\vec{q}}(\omega)=\int_{[0,\vec{q}]}f(T_1^{t_1}T_2^{t_2}\cdots T_d^{t_d}\omega)\,d\vec{t},\ \ S_j=T_j^{q_j},\ \ j=1,...,d.
	$$
	Then, using the calculations from the proof of Theorem 1, we obtain the estimate
	\begin{multline*}
	\sup\limits_{\vec{t}\in\mathbb{R}^d}\left\|\int_{[0,\vec{t}]}f(T_1^{u_1}\cdots T_d^{u_d} \omega)\,d\vec{u}\right\|_p\geq
	\sup\limits_{\vec{t}=\vec{n}\odot\vec{q},\  \vec{n}\in\mathbb{N}^d}\left\|\int_{[0,\vec{t}]}f(T_1^{u_1}\cdots T_d^{u_d} \omega)\,d\vec{u}\right\|_p=
	\\
	=\sup\limits_{\vec{n}\in\mathbb{N}^d}\left\|\int_{[0,\vec{n}\odot\vec{q}]}f(T_1^{u_1}\cdots T_d^{u_d} \omega)\,d\vec{u}\right\|_p=
	\sup\limits_{\vec{n}\in\mathbb{N}^d}\left\|\sum\limits_{0\leq\vec{k}\leq\vec{n}}g_{\vec{q}}(S_1^{k_1}\cdots S_d^{k_d}\omega)\right\|_p.
	\end{multline*}
	
Thus, for commuting automorphisms ${S_j, j=1,...,d}$ and the function $g_{\vec{q}}$, we obtain
$$
\sup\limits_{\vec{n}\in\mathbb{N}^d}\left\|\sum\limits_{0\leq\vec{k}\leq\vec{n}}g_{\vec{q}}(S_1^{k_1}\cdots S_d^{k_d}\omega)\right\|_p\leq C<\infty.
$$
It follows from the work of Bradley~\cite{Brad} that there exists a function ${h_{\vec{q}}\in L_p(\Omega,\mu)}$ such that ${\|h_{\vec{q}}\|_p\leq C}$ and
	${g_{\vec{q}}=\prod_{j=1}^{d}(I-S_j)h_{\vec{q}}}$, i.e., 
	$$
	\int_{[0,\vec{q}]}f(T_1^{u_1}\cdots T_d^{u_d}\omega)\,d\vec{u}=\prod_{i=1}^{d}(I-T^{q_j}_j)h_{\vec{q}},\ \ \vec{q}\in\mathbb{R}^d_+.
	$$
	Let us write this equality for the vector ${\vec{q}/\vec{k}:=\left(\frac{q_1}{k_1},...,\frac{q_d}{k_d}\right)},$ ${\vec{k}\in\mathbb{N}^d}:$
	$$
	\int_{[0,\vec{q}/\vec{k}]}f(T_1^{u_1}\cdots T_d^{u_d}\omega)\,d\vec{u}=\prod_{j=1}^{d}(I-T_j^{q_j \over k_j})h_{\vec{q}/\vec{k}}.
	$$
	Acting on this equality from the left by the operator
	$$
	\prod_{j=1}^d\left(I+T_j^{q_j \over k_j}+T_j^{2{q_j \over k_j}}...+T_j^{(k_j-1){q_j\over k_j}}\right)=\sum_{0\leq\vec{n}\leq\vec{k}}T_1^{n_1{q_1 \over k_1}}\cdots T_d^{n_d{q_d \over k_d}},
	$$ 
	we obtain the equality
	$$
	\int_{[0,\vec{q}]}f(T_1^{u_1}\cdots T_d^{u_d}\omega)\,d\vec{u}=\prod_{j=1}^{d}(I-T_j^{q_j})h_{\vec{q}/\vec{k}}.
	$$
	We will now assume that the vector $\vec{q}$ has binary-rational coordinates. Then there exists a natural number ${m_0=m_0(\vec{q})}$ such that for any ${m\geq m_0}$ we can choose a vector $\vec{k}=\vec{k}(m),$ for which
	$$
	\vec{q}/\vec{k}=(2^{-m},2^{-m},...,2^{-m}):=\vec{v}_m.
	$$
	Indeed, let ${\vec{q}=\left(\frac{a_1}{2^{b_1}},...,\frac{a_d}{2^{b_d}}\right)},$ $ {a_j,b_j\in\mathbb{N}}.$ Then
	$$
	m_0=b_1b_2\cdots b_d,\ \ \vec{k}=(a_12^{m-b_1},a_22^{m-b_2},...,a_d2^{m-b_d}).
	$$
	Since the sequence of functions $h_{\vec{v}_m}$ is bounded, then from the reflexivity of the space ${L_p(\Omega,\mu)}$ for ${p\in(1,\infty)}$ it follows that that there exists a function ${h\in L_p(\Omega,\mu)}$ which is a weak limit of some subsequence $h_{\vec{v}_{m_n}}.$ Then 
	$$
	\prod_{j=1}^{d}(I-T_j^{q_j})h_{\vec{v}_{m_n}}\xrightarrow{\omega}\prod_{j=1}^{d}(I-T_j^{q_j})h.
	$$
	But for ${m_n\geq m_0(\vec{q})}$ there is an equality
	$$
	\prod_{j=1}^{d}(I-T_j^{q_j})h_{\vec{v}_{m_0}}=\int_{[0,\vec{q}]}f(T_1^{u_1}\cdots T_d^{u_d}\omega)\,d\vec{u}=\prod_{j=1}^{d}(I-T_j^{q_j})h_{\vec{v}_{m_n}}.
	$$
	Here we conclude that for any vector $\vec{q}$ with binary-rational coordinates
	$$
	\int_{[0,\vec{q}]}f(T_1^{u_1}\cdots T_d^{u_d}\omega)\,d\vec{u}=\prod_{j=1}^{d}(I-T_j^{q_j})h.
	$$

In the case of the space ${L_1(\Omega,\mu)}$, we use the Komlos theorem (see, for example,~\cite[Theorem~10.10.22]{Bo}), which states the following. For a sequence of functions ${h_{\vec{v}_m}}$ bounded in ${L_1(\Omega,\mu)}$, there exists a subsequence ${h_{\vec{v}_{m_n}}}$ and a function ${h\in L_1(\Omega,\mu)}$ such that there is convergence a.e. of Cesaro means for any of its subsequences $h_{\vec{v}_{m_{n_k}}}$:
$$
\tilde{h}_k:=\frac{h_{\vec{v}_{m_{n_1}}}+h_{\vec{v}_{m_{n_2}}}+...+h_{\vec{v}_{m_{n_k}}}}{k}\to h\ \ \text{as}\ \ \ k\to\infty.
$$	
Then a.e. ${\prod_{j=1}^{d}(I-T_j^{q_j})\tilde{h}_k\to\prod_{j=1}^{d}(I-T_j^{q_j})h}$ as ${k\to\infty}.$ We can assume that ${m_{n_1}\geq m_0(\vec{q})}.$ Then the left-hand side of the last limit relation is equal to
${\int_{[0,\vec{q}]}f(T_1^{u_1}\cdots T_d^{u_d}\omega)\,d\vec{u}}.$
Here we again conclude that for any vector $\vec{q}$ with dyadic rational coordinates
$$
\int_{[0,\vec{q}]}f(T_1^{u_1}\cdots T_d^{u_d}\omega)\,d\vec{u}=\prod_{j=1}^{d}(I-T_j^{q_j})h.
$$	

Thus, the resulting equality is true for all ${p\in[1,+\infty).}$
Since the binary rational numbers are dense in $\mathbb{R},$ and both sides of the last equality are $L_p$-continuous, it will be true for all ${\vec{q}\in\mathbb{R}^d_+}.$
Then
$$
\frac{1}{\pi(\vec{q})}\int_{[0,\vec{q}]}f(T_1^{u_1}\cdots T_d^{u_d}\omega)\,d\vec{u}=\prod_{j=1}^{d}\frac{I-T_j^{q_j}}{q_j}h.
$$
Passing to the limit in $L_p$ for ${q_1,...,q_d\to0},$ taking into account the local ergodic theorem, we obtain ${f=Rh}.$ Which is what was required to be proved.
\end{proof}

\bigskip




\end{document}
